Author's response to reviews

Title: Mortality among patients with tuberculosis requiring intensive care: a retrospective cohort study.

Authors:

Denise R Silva (denise.rossato@terra.com.br)
Diego M Menegotto (diegomenegotto@yahoo.com.br)
Luis F Schulz (lfschulz@uol.com.br)
Marcelo B Gazzana (marcgaz@terra.com.br)
Paulo T.R. Dalcin (pdalcin@terra.com.br)

Version: 3 Date: 28 February 2010

Author's response to reviews: see over
Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is: First name Initial Last name (e.g. Joe F Bloggs)

We changed the authors names, except the first author, because “Silva” is a very common name in Brazil, then we kept the middle and last names.

Tables: we notice that you have included the tables as additional files. If you want the tables to be visible within the final published manuscript please include them in the manuscript in a tables section following the references.

We included the tables following the references.

References - please cite all references in numerical order in the text. Reference 23 does not appear in the correct position.

We correct the references.

Tile page - remove the word Title preceding the main heading, along with the subheadings `Authors names' and `Institution'. Funding information should appear in the Acknowledgements section only.

We removed the word title and the subheadings. We put funding information in the Acknowledgements section.

Abstract - remove keywords.

We removed the keywords.

Highlighting - please remove from the text. You may replace it with bold if you wish.

We removed highlighting.

Authors' contributions: Please indicate that all authors read and approved the final manuscript.

We indicated that all authors read and approved the final manuscript.

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image.

We corrected the figures.
Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

We checked the manuscript for typographical errors.